Author's response to reviews

Title: Prognosis of critically ill cirrhotic versus non-cirrhotic patients: A comprehensive score-matched study

Authors:

Chung-Ming Fu (bugfu0624@gmail.com)
Chih-Hsiang Chang (franwisandsun@gmail.com)
Pei-Chun Fan (franwis1023@gmail.com)
Ming-Hung Tsai (tsccm@ms32.hinet.net)
Shu-Min Lin (smlin100@cgmh.org.tw)
Kuo-Chin Kao (kck0502@cgmh.org.tw)
Ya-Chung Tian (dryctian@cloud.cgmh.org.tw)
Cheng-Chieh Hung (cchung@cloud.cgmh.org.tw)
Ji-Tseng Fang (fangjits@cloud.cgmh.org.tw)
Chih-Wei Yang (cwyang@ms1.hinet.net)
Yung-Chang Chen (cyc2356@adm.cgmh.org.tw)

Version: 3

Date: 19 June 2014

Author's response to reviews: see over
Dear editors and reviewers,

We would like to submit our manuscript entitled “Prognosis of critically ill cirrhotic versus non-cirrhotic patients: A comprehensive score-matched study” by Chung-Ming Fu, Chih-Hsiang Chang, Pei-Chun Fan, Ming-Hung Tsai, Shu-Min Lin, Kuo-Chin Kao, Ya-Chung Tian, Cheng-Chieh Hung, Ji-Tseng Fang, Chih-Wei Yang, Yung-Chang Chen to BMC Anesthesiology for publication as research.

We appreciate very much the comments of the previous reviewers and the editorial office to improve this manuscript. We have amended the manuscript as suggested. The manuscript has been rewritten accordingly and the responses to the comments of the reviewers are described in a point-to-point manner. We hope that the revised manuscript will retain your attention and you will judge the revised manuscript to be suitable for publication in BMC Anesthesiology. We also declare that there is no conflict of interest, support or funding in this manuscript.

Thank you for reviewing our manuscript.

Yours sincerely,

Yung-Chang Chen
Department of Nephrology, Chang Gung Memorial Hospital.
199 Tung Hwa North Road, Taipei, 105 Taiwan
TEL: 886-3-3281200 ext 8181
FAX: 886-3-3282173.
E-mail: cyc2356@adm.cgmh.org.tw